Author’s response to reviews

Title: Experience gained from the implementation of the Saudi TraumA Registry (STAR)

Authors:

Jane Ford (jane.ford@monash.edu)
Abdulrahman Alqahtani (asaqahtani@moh.gov.sa)
Shatha Abuzinada (szinada@ksmc.med.sa)
Peter Cameron (peter.cameron@monash.edu)
Mark Fitzgerald (M.Fitzgerald@alfred.org.au)
Ahmed Alenizi (a.alenizi@ksmc.med.sa)
Dina Farjou (dina.farjou@monash.edu)

Version: 2 Date: 09 Dec 2019

Author’s response to reviews:

10th December 2019

Ms Krisha Mae Chua
JEO Assistant
Journals Editorial Office
Springer Nature

Dear Ms Chua,

Thank you for your email dated Monday 9th December and advice to include Abstract headings in the resubmission of our paper. We are grateful for this further opportunity to comply with the editorial requirements of the journal.
Please note that the abstract includes the four headings as per the instructions to authors at the following link.

https://bmchealthservres.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

These headings are:

• Background
• Methods
• Results
• Conclusions

We trust that this abstract format is acceptable to you. We appreciate your consideration of this resubmission of the manuscript with the inclusion of the required abstract headings.

Yours sincerely,

Jane Ford (Ms)  B.Sc (HIM) CAISS
Senior Registries Manager
Department of Epidemiology and Preventive Medicine
School of Public Health and Preventive Medicine
Monash University
Level 3 553 St Kilda Road
MELBOURNE  3004